\begin{table}%T1
      \caption{Target list.}
         \label{TargetListTable}
%\centerline
{
         \begin{tabular}{llllllllll}
            \hline\hline
            \noalign{\smallskip}
            OBJECT                  & RA$^{a}$ &
            Dec$^{a}$ & $L$ (\lsuno)& Ref.$^{b}$ & $D$ (kpc)& Ref.$^{c}$ & \mj$^{d}$
            & \mh$^{d}$  & \mk$^{d}$  \\
            \noalign{\smallskip}
            \hline
            \noalign{\smallskip}
            \glftn     & 03 01 31.3 & +60 29 13 & \eng{2.1}{4}  & 15,17& 2.20  & 16   & 12.9 &10.0  & 7.7  \\  
            \glfts     & 03 07 24.6 & +58 30 44 & \eng{3.0}{4}  & 18   & 2.70  & 5    & 12.9 &11.3  & 9.8  \\ 
            \glf       & 03 27 38.8 & +58 46 60 & \eng{1.4}{3}  & 7    & 0.90  & 19   & 10.0 & 7.6  & 5.3  \\ 
            \lkhao     & 04 30 14.4 & +35 16 25 & \eng{1.2}{4}  & 7    & 0.80  & 6    &  8.1 & 5.7  & 3.1  \\ 
            \sfour     & 05 40 52.3 & +35 41 31 & \eng{1.2}{3}  & 10   & 1.80  & 9    & 11.0 & 9.9  & 9.0  \\ 
            \sone      & 05 41 06.9 & +35 49 39 & \eng{1.5}{3}  & 10,11& 1.80  & 9    & 12.2 &10.2  & 8.1  \\ 
            \sttft     & 05 41 11.0 & +35 50 02 & \eng{2.5}{3}  & 10   & 1.80  & 9    & 12.2 &8.8   & 6.5  \\
            \ngctwo    & 05 41 45.8 & --01 54 29 & \eng{1.1}{4}  & 21   & 0.40  & 20   & 11.6 &7.4   & 4.6  \\   
            \mont      & 06 07 45.8 & --06 22 54 & \eng{5.0}{3}  & 22   & 0.95  & 1,2  & 18.0 &14.1  & 9.5  \\ 
            \mon       & 06 07 48.1 & --06 22 55 & \eng{6.0}{3}  & 1    & 0.95  & 1,2  & 12.7 & 9.5  & 6.6  \\ 
            \gfive     & 06 08 53.4 & +21 38 29 & \eng{1.1}{4}  & 23   & 1.50  & 23   & 11.6 & 10.7 & 10.3 \\ 
            \stwo      & 06 12 53.9 & +17 59 23 & \eng{3.2}{4}  & 25   & 2.40  & 24   & 16.5 & 12.9 & 9.6  \\ 
            \stwot     & 06 12 54.0 & +17 59 23 & \eng{3.2}{4}  & 25   & 2.40  & 24   & 11.3 & 10.6 & 10.4 \\ 
            \gla-W     & 06 34 37.4 & +04 12 43 & \eng{6.3}{3}  & 26   & 1.60  & 8    & 10.4 & 8.9  & 7.7  \\ 
            \gla-E     & 06 34 37.7 & +04 12 44 & \eng{1.5}{3}  & 26   & 1.60  & 8    & 14.6 & 10.5 & 7.8  \\ 
            \fan       & 06 34 35.6 & +04 12 46 & --            &      & 1.60  & 8    & 13.1 & 11.4 & 9.7  \\ 
            \glb       & 06 41 09.7 & +09 29 36 & \eng{4.0}{3}  & 27   & 0.75  & 14   & 11.0 & 7.6  & 4.8  \\ 
            \sir       & 20 27 26.8 & +37 22 48 & \eng{2.0}{4}  & 29   & 0.60  & 28   & 10.4 & 7.7  & 5.9  \\ 
            \glt       & 20 29 24.9 & +40 11 20 & \eng{4.0}{4}  & 30   & 1.50  & 30   & 14.4 &11.0  & 6.1  \\ 
%           \ws        & 20 38 36.4 & +42 37 35 & \eng{1.0}{4}  &      & 2.00  &      & 14.0 &11.3  & 9.2  \\ 
            \sonef     & 22 19 18.1 & +63 18 47 & \eng{5.0}{3}  & 3    & 0.90  & 4    & 11.7 & 8.6  & 6.2  \\
            \ngcsfte   & 23 13 45.4 & +61 28 12 & \eng{5.0}{4}  & 12   & 2.70  & 13   & 15.3 &14.1  & 8.9  \\
            \hline
         \end{tabular}}
\par
         \medskip
      $^{{a}}$ Equinox J2000.\\ 
      $^{{b}}$ References for the luminosity.\\ 
      $^{{c}}$ References	for the distance.\\ 
      $^{{d}}$ Magnitudes extracted from the literature.\\ 
      Key to the references: 1. \citet{Beckwith76};
       2. \citet{Thronson80a}; 3. \citet{Mozurkewich86};
     4. \citet{Crampton74}; 5.~\citet{CohenKuhi77};
     6. \citet{Herbig71}; 7. \citet{Harvey79a}; 8. \citet{Blitz80};
     9. \citet{EvansBlair81}; 10. \citet{Evans81};
     11. \citet{Nordh84}; 12. \citet{CampbellThompson84}; 
     13. \citet{Blitz82}; 14. \citet{Park00}; 15. \citet{Snell88};
     16.~\citet{BeckerFenkart71}; 17. \citet{Thronson80b};
     18. \citet{Wynn-Williams81}; 19. \citet{Harvey79b};
     20. \citet{AnthonyTwarog82}; 21. \citet{Jiang84};
     22. \citet{Howard94}; 23. \citet{Wu96}; 24. \citet{Itoh01};
     25. \citet{Howard97}; 26. \citet{Castelaz85};
     27. \citet{Harvey77}; 28. \citet{Staude82};
     29. \citet{Harvey82}; 30. \citet{WynnWilliams82}.
   \end{table}